Author’s response to reviews

Title: Establishment of Rat Ankle Post-Traumatic Osteoarthritis Model Induced by Malleolus Fracture

Authors:
Dawei Liang (527079901@qq.com)
Jian Sun (13403415163@163.com)
Fangyuan Wei (footwfy@126.com)
Jianzhong Zhang (trfoot@126.com)
Pengcui Li (lpc1977@163.com)
Yingke Xu (Yxu11@tulane.edu)
Xianwen Shang (shangxianwen@medmail.com.cn)
Jin Deng (742717116@qq.com)
Ting Zhao (tzhao@bu.edu)
Lei Wei (lei_wei@brown.edu)

Version: 2 Date: 03 Aug 2017

Author’s response to reviews:

Re: BMC Musculoskeletal Disorders: BMSD-D-17-00460R1

Title: Establishment of Rat Ankle Post-Traumatic Osteoarthritis Model Induced by Malleolus Fracture

Authors: Dawei Liang; Jian Sun; Fangyuan Wei; Jianzhong Zhang; Pengcui Li; Yingke Xu; Xianwen Shang; Jin Deng; Ting Zhao; Lei Wei

Dear Prof. Ahrens,

Thank you very much for your reconsidering our revised manuscript for potential publication in BMC Musculoskeletal Disorders. Enclosed you will find the revised version of our manuscript.
According to your comments related to the language improvement, the revised manuscript has been carefully edited by a native and professional editor, Dr. Ericka M. Bueno. We tried our best to improve the manuscript and we appreciate for your warm work earnestly.

We are looking forward to hearing from you soon.

Sincerely,

Lei Wei, MD., Ph.D.
Associate Professor
Department of Orthopedics,
Brown Medical School/Rhode Island Hospital,
Suite 402A, 1 Hoppin Street,
Providence RI 02903